Author's response to reviews

Title: How Much Dentists are Ethically Concerned about Overtreatment; a Vignette-Based Survey in Switzerland

Authors:

Ali Kazemian (Ali.Kazemian@usb.ch)
Isabelle Berg (isabelle.berg@usb.ch)
Christina Finkel (cristinafinkel@yahoo.com)
Shahram Yazdani (shahram.yazdani@yahoo.com)
Hans-Florian Zeilhofer (hfzeilhofer@uhbs.ch)
Philipp Juergens (pjuergens@uhbs.ch)
Stella Reiter-Theil (s.reiter-theil@unibas.ch)

Version: 3 Date: 27 November 2014

Dear editor,

In response to your message regarding the necessity of additional modification for the manuscript #1371378811144856, "How Much Dentists are Ethically Concerned about Overtreatment; a Vignette-Based Survey in Switzerland”, we hereby kindly submit our revised manuscript.

A paragraph has been added to the 8th page of the MS (line numbers 139-143) concerning the approval of the ethics committee and the issue of obtaining informed consent.

Please let us know if any other modification was wanted by the reviewers. We hope you find our manuscript suitable for publication in the journal of BMC Medical Ethics and look forward to hearing from you.

Sincerely,

Ali Kazemian
DDS. PhD.
Department of Community Oral Health, Faculty of Dentistry, Mashhad University of Medical Sciences, Mashhad, Iran.
00989155033050
kazemiana@mums.ac.ir, Ali.Kazemian@usb.ch